Author's response to reviews

Title: Gene expression profiling identifies activated growth factor signaling in poor prognosis (Luminal-B) estrogen receptor positive breast cancer

Authors:

Sherene Loi (sherene.loi@bordet.be)
Christos Sotiriou (christos.sotiriou@bordet.be)
Benjamin Haibe-Kains (bhaibeka@ulb.ac.be)
Francoise Lallemand (francoise.lallemand@bordet.be)
Nelly M Conus (nelly.conus@petermac.org)
Martine J Piccart (martine.piccart@bordet.be)
Terence P Speed (terry@wehi.edu.au)
Grant A McArthur (grant.mcarthur@petermac.org)

Version: 3 Date: 5 May 2009

Author's response to reviews:

References were unformatted in previous "clean" version. My apologies.